Check for updates

Citation: Pershina AG, Ivanov VV, Efimova LV, Shevelev OB, Vtorushin SV, Perevozchikova TV, et al. (2017) Magnetic resonance imaging and spectroscopy for differential assessment of liver abnormalities induced by *Opisthorchis felineus* in an animal model. PLoS Negl Trop Dis 11(7): e0005778. https://doi.org/10.1371/journal. pntd.0005778

Editor: jong-Yil Chai, Seoul National University College of Medicine, REPUBLIC OF KOREA

Received: March 31, 2017

Accepted: July 5, 2017

Published: July 14, 2017

Copyright: © 2017 Pershina et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This work was supported by the Russian Science Foundation (http://www.rscf.ru/en) [grant number 14-15-00247 to AGP, AES, LMO], and the Russian Foundation for Basic Research (http:// www.rfbr.ru/rffi/eng) [grant number 15-04-05580 a to VVI]. Animal MRI/MRS was conducted at the RESEARCH ARTICLE

Magnetic resonance imaging and spectroscopy for differential assessment of liver abnormalities induced by *Opisthorchis felineus* in an animal model

Alexandra G. Pershina^{1,2}*, Vladimir V. Ivanov¹, Lina V. Efimova¹, Oleg B. Shevelev³, Sergey V. Vtorushin¹, Tatjana V. Perevozchikova¹, Alexey E. Sazonov¹, Ludmila M. Ogorodova¹

 Central Research Laboratory, Siberian State Medical University, Tomsk, Russia, 2 Department of Biotechnology and Organic Chemistry, National Research Tomsk Polytechnic University, Tomsk, Russia, 3 Center for Genetic Resources of Laboratory Animals, Institute of Cytology and Genetics SB RAS, Novosibirsk, Russia

* allysyz@mail.ru

Abstract

Background

European liver fluke *Opisthorchis felineus*, causing opisthorchiasis disease, is widespread in Russia, Ukraine, Kazakhstan and sporadically detected in the EU countries. *O. felineus* infection leads to hepatobiliary pathological changes, cholangitis, fibrosis and, in severe cases, malignant transformation of bile ducts. Due to absence of specific symptoms, the infection is frequently neglected for a long period. The association of opisthorchiasis with almost incurable bile duct cancer and rising international migration of people that increases the risk of the parasitic etiology of liver fibrosis in non-endemic regions determine high demand for development of approaches to opisthorchiasis detection.

Methodology/Principal findings

In vivo magnetic resonance imaging and spectroscopy (MRI and MRS) were applied for differential assessment of hepatic abnormalities induced by *O. felineus* in an experimental animal model. Correlations of the MR-findings with the histological data as well as the data of the biochemical analysis of liver tissue were found. MRI provides valuable information about the severity of liver impairments induced by opisthorchiasis. An MR image of *O. felineus* infected liver has a characteristic pattern that differs from that of closely related liver fluke infections. 1H and 31P MRS in combination with biochemical analysis data showed that *O. felineus* infection disturbed hepatic metabolism of the host, which was accompanied by cholesterol accumulation in the liver.

Conclusions

A non-invasive approach based on the magnetic resonance technique is very advantageous and may be successfully used not only for diagnosing and evaluating liver damage induced Centre for Genetic Resources of Animal Laboratory at the Institute of Cytology and Genetics, Siberian Branch, Russian Academy of Sciences (Ministry of Education and Science of the Russian Federation (http://www.fcpir.ru/) projects Nos.

RFMEFI61914X0005 and RFMEFI62114X0010). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

by *O. felineus*, but also for investigating metabolic changes arising in the infected organ. Since damages induced by the liver fluke take place in different liver lobes, MRI has the potential to overcome liver biopsy sampling variability that limits predictive validity of biopsy analysis for staging liver fluke-induced fibrosis.

Author summary

Opisthorchiasis caused by Opisthorchis felineus is a fish-borne parasitic worm infection spread in Russia and some European countries. The morbidities provoked by O. felineus infection are cholangitis and bile duct fibrosis. Long-term infection is associated with high risk of developing cholangiocarcinoma, a generally incurable bile duct cancer type. Due to lack of specific symptoms, O. felineus infection often escapes detection. Thus, opisthorchiasis diagnosis, especially in non-endemic regions, is a serious problem for physicians. In the paper, an animal model of O. felineus induced opisthorchiasis has been evaluated by in vivo magnetic resonance spectroscopy (MRS) and magnetic resonance imaging (MRI). Application of MR-techniques allowed to detect a characteristic MRI pattern of O. felineus infected liver, get valuable information about the severity of organ impairments, and bring to light some metabolic changes provoked by the helminth in the liver. Since damages take place in different liver lobes, MRI has the potential to overcome sampling variability of liver biopsy that limits liver fluke-induced fibrosis staging. The use of MRtechniques is very advantageous for investigating parasitic infection. Collection of experimental MR-data gives a new impulse to examination of infected humans and encourages to implement these methods in routine diagnosis of infections, including but not limited to opisthorchiasis.

Introduction

Two liver fluke species of the genus *Opisthorchis—O. felineus* and *O. viverrini*—are known as human pathogenic agents. This fluke parasite inhabits the bile duct of the host liver, causing local mechanical damage and chronic inflammation, and having general toxic effects on the whole body [1]. Bile duct dilatation and mechanical obstruction accompanied by bile sludge formation are common complications of chronic opisthorchiasis [2]. Generally, liver fluke infection causes hepatobiliary pathological changes and leads to cholangitis, cholecystitis, and cholelithiasis.

The most severe complication of opisthorchiasis is malignant transformation. There is a significant association between cholangiocarcinoma (CAA) and *O. viverrini* infection; that liver fluke is considered carcinogenic (group I) to humans [3–5]. *O. felineus* is classified by IARC as group 3 due to limited experimental data [6]. However, in sources published in the Russian language, the association between CAA and chronic long-term *O. felineus* infection has been emphasized [7,8]. In whole, periductal fibrosis is considered an important risk factor for bile duct cancer and is associated with CCA development [9,10]. In the meantime, there is no significant association between liver fluke infection and cirrhosis, at least for closely related species *O. viverrini* and *Clonorchis sininsis* [5].

Chronic liver diseases, mainly liver fibrosis, represent a public health problem worldwide [11]. The generally discussed disease entities associated with liver fibrosis are nonalcoholic fatty liver disease (NAFLD), alcoholic liver disease, chronic viral hepatitis and chronic

cholestasis, primary biliary cholangitis, and primary sclerosis cholangitis [12]. Fibrosis induced by liver flukes is commonly out of focus, though chronic infection caused by helminths of the genus *Opisthorchis* leads to periductal fibrosis and severe abnormalities in the hepatobiliary system [3].

It is clear that the pathogenesis of liver fibrosis depends on the underling etiology [13]. The main reason for complicated recognition of the cause of fibrotic or inflammatory process consists in the fact that liver responds to various injuries in a limited number of ways [14].

Parasitic liver flukes of the genus *Opisthorchis* are widespread in Eurasia—in Russia (mainly West Siberia), Ukraine, Kazakhstan (*O. felineus*), and South Asia (*O. viverrini*). In the endemic regions more than 750 million people are at high risk of liver fluke infection [5]. There are some reports about *O. felineus* infected people in EU countries [15,16]. It is very important to note that most infected persons, especially those with chronic infection (liver fluke lifespan may exceed 25 years [4]), show no symptoms at all or non-specific symptoms which are very difficult to recognize, so the infection is frequently neglected [17,18]. Taking into consideration increasing international migration of people, it is essential to bear in mind the risk of the parasitic etiology of liver fibrosis in the non-endemic regions [19]. In time, non-detected opisthorchiasis infection may be a cause of false diagnosis and complications in the treatment strategy of choice [17]. Furthermore, it is very important to exclude liver fluke infection, when assessing the transplantation potential of living liver donors.

Search for non-invasive tests for liver fibrosis diagnosis is the main trend in hepatology, since the "gold-standard" method—liver biopsy—has some shortcomings, such as invasiveness, complications, and sampling variability [11,13,20]. MR imaging-based techniques for liver fibrosis assessment are very promising and are being actively developed currently [21].

Magnetic resonance imaging (MRI) and magnetic resonance spectroscopy (MRS) share common physical background and are based on a nuclear magnetic resonance (NMR) phenomenon [22]. In the presence of a static magnetic field, for a nucleus of spin 1/2, absorption of energy emitted by a radiofrequency coil induces "flips" of a magnetic moment to another orientation. This "spin flip" places some of the spins in their high energy state. Relaxation processes which return nuclei back to the lower energy state after switching off the radiofrequency signal come with generation of a measurable amount of the radio frequency signal. The produced NMR signals received by the radiofrequency coil allow to generate MRS and MRI after Fourier transformation.

MRI has become an increasingly important imaging technique for investigating patients with hepatic and biliary disorders [23,24]. Although MRI examination is more expensive in comparison with computed tomography and ultrasound, this modality is widely used due to higher spatial resolution [25] and absence of ionizing radiation. MRS methods offer an opportunity to assess relative tissue metabolite concentrations based on the chemical shift phenomenon [26] and are very useful in clinical and biomedical studies for examining metabolic changes *in vivo* non-invasively [25,27]. In particular, 1H MRS is successfully used to determine relative lipid concentration in hepatic tissue [28], whereas 31P MRS allows to detect phosphorylated metabolites, including high-energy phosphates [25]. MRS methods are implemented to investigate the hepatic metabolic state and are applied as non-invasive diagnostic techniques for studying hepatobiliary pathology [29–31].

Despite the fact that MRI and MRS have become increasingly important imaging techniques for investigation of patients with liver and biliary disease [23,24], there are few papers describing application of *in vivo* MR-techniques for studying liver fluke infection [32–35] and no papers depicting the use of these methods for investigating *O. felineus*.

In this study we used magnetic resonance techniques (MRI/MRS) for differential assessment of liver abnormalities induced by *O. felineus* in an experimental animal model. We also found out correlations between the MR-findings and the histological data as well as the data of the biochemical analysis of liver tissue.

Materials and methods

Ethics statement

The experimental protocol was approved by the Bioethics Review Committee of the Institute of Cytology and Genetics, Siberian Branch, Russian Academy of Sciences, Novosibirsk (No. 30 from 20.11.2015). All animal experiments were conducted according to the principles of the Guide for the care and use of laboratory animals [36].

Experimental opisthorchiasis model

Metacercariae of *O. felineus* were obtained from naturally infected fish (*Leuciscus leuciscus*) caught in the river in the endemic areas of Western Siberia (the Tom, Tomsk), Russia. A permit was not required to collect these fish according Federal law of the Russian Federation No. 166 "About fisheries and conservation of water biological resources" from 20.12.2004. For the experiment, 5-week-old male hamsters (n = 8) (*Mesocricetus auratus*) were infected intragastrically with 50 metacercariae per hamster, according to the previously described protocol [37]. Age-matched intact male hamsters (n = 8) were used as controls. The hamsters were housed two in a cage (OptiRAT) under conventional conditions and were permitted ad libitum access to food and water. The animals were handled in pathogen-free environment. At 8 weeks post-infection, the hamsters were scanned with MRI and MRS *in vivo*. Following the MR examinations, the subsets of the infected and intact animals were deeply anesthetized with carbon dioxide and euthanized by decapitation. Blood and liver samples were collected for examination from each hamster from the control (n = 8) and infected (n = 8) groups. Serum and blood analyses were performed using routine procedures (Methods in S1 Text).

MRI and MRS in vivo

All 1H/31P MR experiments were performed on a horizontal tomographic scanner with magnetic field intensity of 11.7 T (Bruker, Biospec 117/16 USR, Germany). Prior to MR examinations, the animals were fasted overnight. The animals were anaesthetized with gas anesthesia (Isofluran; Baxter Healthcare Corp., Deerfield, IL) using a Univentor 400 Anesthesia Unit (Univentor, Zejtun, Malta). The animal body temperature was maintained with a water circuit installed into the table bed of the tomographic scanner, which maintained the temperature of 30°C on its surface. A pneumatic respiration sensor (SA Instruments, Stony Brook, NY) was placed under the lower body part, which allowed to control the anesthesia depth.

MRI. All MR images were recorded with a receiver—transmitter 1H volume coil (T11440V3). High-resolution T2-weighted images of the hamster liver (section thickness 1 mm; field of vision (FOV) 4.0×4.0 mm; matrix 512×512 dots) were acquired with respiratory triggering using TurboRARE (Rapid Imaging with Refocused Echoes) method with pulse sequence parameters: TE_{eff} = 18 ms, TR = 900 ms, Flip Angle = 180°, RARE Factor = 4. Serial images in axial and coronal orientation were recorded for each animal.

1H/31P-MRS. Proton and phosphorus spectra were recorded with a receiver—transmitter double-tuned 1H/31P surface coil (T11619V3). The coil was positioned to the right upper abdominal quadrant of the hamster liver in the right lateral decubitus position, which allowed to minimize the signal capture by muscles and decrease movement artifacts. Before each session, recording of 3 orthogonal packages of liver sections (Multi-slice Tri-Pilot scanning) was conducted, to guide the correct coil positioning relative to liver tissue (in case of infected

hamster to capture liver damaged zone). The protons spectra were recorded using the spatially localized PRESS (Point Resolved Spectroscopy) method with TE = 20 ms, TR = 2.5 ms, and 256 replicates, voxel size $5\times5\times5$ mm. Water signal in the spectra was suppressed by the variable power and optimized relaxation delays (VAPOR) method.

The phosphorus spectra were recorded using the non-spatially localized single-pulse method with TE = 3 ms, TR = 100 ms, and 2,048 replicates. The spectrum range was 40 ppm, which made it possible to obtain information about the tissue contents of 5 substances represented by 7 peaks, phosphomonoester (PME; 6 ppm), inorganic phosphate (Pi; 5 ppm), phosphodiester (PDE; 2 ppm), creatine phosphate (PCr; 0 ppm originated from muscle contamination), and nucleoside triphosphate (NTP; 2.7, 7.8, and 16.5 ppm).

MRS data processing. Time domain data were Fourier transformed after Gaussian multiplication (center, 0 ms; width, 30 ms) and phase corrected. Quantification of the spectral peak areas was performed using the TOPSPIN 2.0 PV software package (Brucker), including polynomial baseline correction followed by frequency domain curve peak integration. The 1H and 31P metabolite concentrations were calculated from the peak areas and expressed relative to an area under the curve. Intracellular pH was calculated based on the chemical shift difference δ between Pi and α -NTP ($\delta = f_{Pi}$ - $f_{\alpha NTP}$ -7.56) using the Henderson-Hasselbalch equation [38]:

$$pHi = 6.75 + \log 10 [(\delta Pi - 3.27) / (5.69 - \delta Pi)]$$
(1)

Histological analysis

Liver samples (from posterior segments of the right lobe) were fixed in 10% buffered formalin, and embedded in paraffin. Tissue sections were cut into $4-5 \mu$ m-thick slices and stained with hematoxylin and eosin. Histological analysis was performed with the optical microscope Axiostar plus (Carl Zeiss, Germany). For differential staining of collagen in the liver samples, Van Gieson's method was used. Fibrosis was graded according to METAVIR score [39]. Hepatic steatosis was graded on a 0–3 scale through visual estimation of the percentage of hepatocytes containing intracellular vacuoles of fat [40].

Liver tissue analysis

Two samples of liver tissue were taken from posterior segments of the right lobe and medial segments of the left lobe for each animal. The samples were divided into five pieces in accordance with subsequent examinations and immediately placed in liquid nitrogen. Homogenization was performed using Tissue Grinders (PELLET PESTLE[®] Cordless Motor, Kimble Chase, TN) on ice.

Lipids assay. Extraction of lipids from 200 mg of liver tissue (two pieces for each animal) was performed using Folch method [41]. The concentrations of cholesterol, triglycerides and phospholipids in the lipid extract were determined spectrophotometrically (SF2000, Russia) using Chronolab kits (Spain). The total lipids were gravimetrically determined by drying 2 ml of liver lipid extract in a glass vial.

ATP assay. 60 mg of liver tissue (two pieces for each animal) was homogenized in 1 ml of 3% trichloroacetic acid and centrifuged at 4,000 g, 4°C, 20 min. The pH of the supernatant was adjusted to 7.4 with 1 M Tris. The ATP concentration was measured on Anthos lucy2 (Asys Hitech GmbH, Austria) using the Adenosine 5'-triphosphate (ATP) Bioluminescent Assay Kit (Sigma-Aldrich, St. Louis, MO).

Protein assay. 30 mg of liver tissue (two pieces for each animal) was homogenized in 400 µl of NP-40 buffer (150 mM sodium chloride, 1.0% Triton X-100, 50 mM Tris, pH 8.0) and centrifuged at 12,000 g, 4°C, 20 min. Then the protein concentration was determined in

the supernatant using Bradford assay [42]. Bovine serum albumin (BSA fraction V, Sigma) was used as a protein standard.

Glycogen assay. 20 mg of liver tissue (two pieces for each animal) was homogenized in 1 ml of 0.3 M HClO₄. The homogenate was incubated with aminoglucosidase (AG, Sigma) in 50 mM sodium acetate, 0.02% BSA, pH 5.5 for 2 h at RT, and centrifuged at 10,000 g for 1 min. The amount of the released glucose was determined by the Hexokinase Method [43] using the Glucose Reagent (Vector-Best, Russia). Glucose (Sigma) was used as a standard. Free glucose in the homogenate was measured without addition of AG and subtracted from AG-treated samples.

Western blot. 20 mg of liver tissue was homogenized on ice using dounce homogenizer in 1 ml of lysis buffer (50 mM Tris-HCl (pH 7.5), 250 mM sucrose, 5 mM sodium pyrophosphate (NaPPi), 50 mM NaF, 1 mM EDTA, 1 mM dithiothreitol (DTT), 0.5 mM phenylmethylsulfonyl fluoride (PMSF)) supplemented with Mammalian Protease inhibitor Cocktail (P8340 Sigma). The protein concentration was quantified by Bradford assay, using BSA (Sigma) as a standard. Following lysis sodium dodecyl sulfate (SDS) was added to a final concentration of 0.2% and samples are boiled for 5 min. The samples (twenty micrograms of total proteins) were loaded onto a 15% SDS-PAGE gel according to Laemmli protocol [44]. The separated proteins were transferred to nitrocellulose membranes. The membranes were blocked with 3% non-fat milk in Tris-buffered saline (TBS) with 0.1% Tween 20 and 50 mM NaF for 1 h at room temperature and probed overnight at 4°C with primary antibodies in 5% BSA supplemented with 50 mM NaF, followed by incubation with Goat Anti-Rabbit IgG H&L (HRP) (ab6721) (Abcam, USA) in TBS supplemented with 5% milk and 50 mM NaF for 1 h at room temperature. The membranes were developed with a ECL Chemiluminescent Substrate Reagent Kit (Thermo Scientific, USA) and scanned by G:box XT4 (Syngene). The band intensities were quantified using GeneTools (Synegene). The primary antibodies Phospho-AMPKa (Thr172) (CellSignaling technology, Inc.) and Anti-beta Actin antibody (ab8227) (Abcam, USA) were used.

Statistical analysis

Statistical analyses were performed using IBM SPSS Statistics for Windows, Version 21.0 (Armonk, NY: IBM Corp.). Data were presented as median (range) for data with non-normal distribution and mean (SD) for data with normal distribution. The data were tested by the Shapiro—Wilk test for normality. The differences among continuous variables with normal distribution were analyzed by the t-test and among continuous variables with non-normal distribution—by the Mann-Whitney test. For correlation analyses, the Person (for continuous variables) and Spearman (for ordinal variables) correlations were used. The P value below 0.05 was considered as significant.

Results

Histological analysis

Histological analysis data showed that *O. felineus* infection resulted in derangement of the liver architecture, whereas the trabecular pattern mostly remained intact. Hepatocytes had various sizes and patchy exhibited cloudy swelling. Cholangitis in the portal tract was accompanied by pronounced periductal and mild portal fibrosis as well as focal cystic dilatations of the intrahepatic bile ducts. Small bile ductular proliferation and irregular periportal (mostly moderate) infiltration of inflammatory cells, with predominant population of lymphocytes and histiocytes, occurred (Fig 1). Fibrosis was irregular; patchy, incomplete, thin fibrous septa (portal to portal and portal to central bridge) were observed (Fig 1A, 1B and 1C). In whole, chronic



Fig 1. Histological analysis of hamster liver. (A) Microphotographs of *Opisthorchis felineus* infected hamster liver at 8 weeks post-infection. An adult fluke is found in bile duct lumen. Small bile duct proliferation and chronic cholangitis accompanied by inflammatory infiltration in the bile duct wall; hematoxylin-eosin staining, ×100. (B) Mild portal fibrosis with thin septa formation (portal to portal and portal to central) and (C) pronounced periductal fibrosis with chronic inflammation around the bile duct are observed. In adjacent hepatic tissue, fibrosis with porto-portal septa formation takes place. Van Gieson's staining showing extensive deposition of mature collagen fibers, ×100. (D) Microphotographs of the uninfected hamster liver. No pathological changes are observed. Hematoxylin-eosin staining, ×100 and (E) Van Gieson's staining, ×100.

https://doi.org/10.1371/journal.pntd.0005778.g001

cholangitis (mild to moderate inflammatory activity, A1-A2 according to METAVIR score) and mild chronic hepatitis were diagnosed, and the stage of fibrosis varied from 1 to 2 according to METAVIR score. There were no lipid droplets in the tested liver samples. No pathological changes took place in the liver in the reference group (Fig 1D and 1E).

Blood and serum analysis

The results of blood and serum analyses are given in Results in S1 Text and Table in S1 Table. In the infected group of hamsters the levels of alanine aminotransferase (ALT) and gamma-glutamyl transferase (GGT) increased markedly. Significant elevation of cholesterol, triglycer-ides, and low-density lipoproteins (LDL) in serum of the infected animals was detected. All the above listed serum parameters correlated with the fibrosis stage. The concentration of albumin in serum of the infected hamsters was lower than in the reference group and was accompanied by a statistically significant increase in the urea concentration. However, there was no correlation between these parameters (r = -0.271, p = 0.309).

Biochemical analysis of liver tissue

The liver to body index increased in the infected animals, while the spleen to body index remained unchanged (S1 Table). The results of liver tissue biochemical analysis are given in Table 1. There was a statistically significant rise in the cholesterol level in opisthorchiasis, and the cholesterol to phospholipid ratio also elevated in the infected liver. Total lipid content, triglycerides and phospholipids as well as glycogen did not differ between the control and infected groups of hamsters. A significant decrease in the protein concentration occurred in the infected livers. However, the calculated total protein content in the entire liver did not differ between both control and experimental groups due to enlargement of the organ in the infected hamsters. It is important to note, that infection did not lead to lowering of the ATP concentration in liver tissue.

Table 1. Liver tissue biochemical analysis.

Metabolite [#]	Con	trol	Infec	cted	Correlation with fibrosis stage		
	mean	SD	mean	SD	r	p-level	
Protein, mg g ⁻¹	120.80	17.19	94.92*	15.62	-0.720	0.002	
Total cholesterol, mg g ⁻¹	1.80	0.25	2.64*	0.59	0.781	0.000	
Triglycerides, mg g ⁻¹	2.84	0.39	3.29	1.23	0.282	0.290	
Phospholipids, mg g ⁻¹	9.89	0.84	8.93	1.05	-0.447	0.082	
Total lipids, mg g ⁻¹	42.77	4.91	39.88	4.38	-0.136	0.615	
Triglycerides to phospholipid ratio	0.26	0.04	0.32	0.09	0.463	0.071	
Cholesterol to phospholipid ratio	0.37	0.06	0.60*	0.10	0.885	0.000	
ATP, nmol g ⁻¹	2.77	0.66	2.94	1.04	-0.032	0.905	
Glycogen, mg g ⁻¹	4.17	3.41	6.11	2.16	0.152	0.573	

*Concentration calculated per gram wet weight of tissue

*difference between the infected and the control group is significant at p<0.01

r, correlation coefficient; ATP, Adenosine triphosphate.

https://doi.org/10.1371/journal.pntd.0005778.t001

¹H MRI

A T2-weighted image of hamster liver infected with *O. felineus* for 8 weeks displayed pronounced bile duct abnormalities (Fig 2A, 2B and 2C). Hyperintense areas extended around the bile duct are found in different lobes of the liver in all infected for 8 weeks hamsters. The most common MR findings included periductal enhancement, intrahepatic bile duct dilatation, and



Fig 2. MRI of hamster liver. T2-weighted (A) axial and (B, C) coronal images of *O. felineus* infected liver and (D) axial and (E, F) coronal images of uninfected hamster liver (control) at 11.7 T. TurboRARE (Rapid Imaging with Refocused Echoes) with respiratory triggering, with pulse sequence parameters $TE_{eff} = 18 \text{ ms}$ and TR = 900 ms, Flip Angle = 180° , RARE Factor = 4. Bruker, Biospec 117/16 USR, Germany. In *O. felineus* infected hamster liver, the hyperintense signals from the bile duct wall as well as the bright zones extending from the dilated bile duct due to inflammation and fibrosis are clearly depicted. There are no pathological changes in the liver of the control hamsters.

https://doi.org/10.1371/journal.pntd.0005778.g002



Fig 3. Typical *in vivo* **1H** magnetic resonance spectrum of hamster liver. Spectrum was recorded using the spatially localized PRESS (Point Resolved Spectroscopy) method with TE = 20 ms, TR = 2.5 ms, 256 replicates, voxel size $5 \times 5 \times 5$ mm. Water signal in the spectra was suppressed by the variable power and optimized relaxation delays (VAPOR) method. Bruker, Biospec 117/16 USR, Germany. Lipid resonance peaks are labeled: P1: methyl proton CH₃ (0.9 ppm); P2: methylene ($-CH_2-$)_n (1.3 ppm); P3: $-CH_2-CH = CH-$ (2.0 ppm); P4: $-CH_2-COO-$ (2.2 ppm); P5: $-CH = CH-CH_2-CH = CH-$ (2.8 ppm).

https://doi.org/10.1371/journal.pntd.0005778.g003

bile duct wall thickening. We detected a hyperintense zone extending from the dilated bile duct due to inflammation and fibrosis, without pronounced peripheral bile duct enhancement. In the control group the liver parenchyma was homogenous and showed no dilatation of the bile ducts (Fig 2D, 2E and 2F).

1H MRS

Analysis of the proton spectra (Fig 3) showed no statistically significant differences in the content of lipid compounds in the livers of the control and infected hamster groups (Table 2).

We found no correlation between the concentration of lipids in liver tissue determined according to biochemical analysis and certain characteristic peak areas in *in vivo* 1H MRS spectra. While the triglyceride and cholesterol concentrations, measured by biochemical analysis of liver tissue, correlated with 2.2 to 0.9 ppm peak ratio (r = 0.561, p = 0.041 for both), the triglyceride level correlated with 2.8 to 0.9 ppm peak ratio (r = 0.537, p = 0.032).

In fact, a peak at 0.9 ppm is attributed to protons in the methyl group of cholesterol, phospholipids and triglycerides, whereas peaks at 2.2 ppm and 2.8 ppm are assigned to protons bonded with carbons alpha and polyunsaturated carbons of triglycerides [45] [46]. More specifically 2.8 ppm/0.9 ppm peak area ratio is associated with a polyunsaturated bond [47,48].

	<u>`</u>	,					
	0.9 ppm	1.3 ppm	2.0 ppm	2.2 ppm	2.8 ppm	3.2 ppm	5.3 ppm
Infected	0.09	0.73	0.08	0.05	0.02	0.02	0.01
SD	0.03	0.09	0.03	0.04	0.02	0.02	0.01
Control	0.10	0.73	0.08	0.04	0.02	0.01	0.01
SD	0.03	0.06	0.02	0.02	0.01	0.02	0.02
Correlation coefficient with fibrosis stage	-0.250	0.026	0.062	0.042	0.041	-0.038	0.041
p-level	0.351	0.924	0.821	0.877	0.881	0.889	0.880

Table 2. Levels of hepatic lipid metabolites (in vivo 1H MRS).

https://doi.org/10.1371/journal.pntd.0005778.t002



Fig 4. Typical *in vivo* hepatic 31P magnetic resonance spectrum of the right upper abdominal quadrant (includes liver, abdominal wall, subcutis, etc.) of a hamster. Spectrum was recorded using the nonspatially localized, single-pulse method with TE = 3 ms, TR = 100 ms, 1048 replicates. Bruker, Biospec 117/16 USR, Germany. Phosphorylated metabolites resonance peaks are labeled: P1: phosphomonoester (PME); P2: inorganic phosphate (Pi); P3: phosphodiester (PDE); P4: creatine phosphate (PCr); P5: NTP- γ ; P6: NTP- α ; P7: NTP- β .

https://doi.org/10.1371/journal.pntd.0005778.g004

Thus, according to the 1H MRS study, there were no significant changes in the lipid profile in the liver of the infected hamsters.

31P MRS

Fig 4 presents typical 31P MR spectra of the hamster liver. The calculated fractions of various phosphorylated metabolites are given in Table 3. Analysis of phosphorous spectra showed that the hepatic PME level increased in the infected group of hamsters, while the PDE levels remained unchanged. Significantly lower levels of NTP- α were detected in the infected group of hamsters, whereas the peak area attributed to other high-energy compounds, such as NTP- β and NTP- γ , in the livers of the control and infected hamsters had no statistically significant differences. PME and NTP- α peak areas correlated with the fibrosis stage (r = 0.710, p = 0.002 and 0.726, p = 0.001, respectively).

Correlation analysis revealed that PME and NTP- α peak areas significantly correlated with serum ALT, AST, cholesterol, HDL, cholesterol to phospholipids ratio in liver tissue as well as with the fibrosis stage (S2 Table).

Table 3. Levels o	f hepatic phosp	horylated metabolites	(in vivo 31P MRS).
-------------------	-----------------	-----------------------	--------------------

	PME	Pi	PDE	ΝΤΡγ	ΝΤΡα	ΝΤΡβ	рН _і	ΝΤΡγ/ ΝΤΡα	ΝΤΡγ/ ΝΤΡβ	ΝΤΡβ/ ΝΤΡα	PME/PDE	Pi/NTPα	Ρί/ΝΤΡβ	Ρί/ΝΤΡγ
Infected	0.17**	0.09	0.04	0.11	0.24**	0.17	7.22	0.47	0.71	0.70*	4.45	0.38**	0.58	0.82*
SD	0.03	0.02	0.01	0.02	0.02	0.03	0.11	0.08	0.31	0.13	1.38	0.09	0.28	0.14
Control	0.13	0.08	0.04	0.12	0.28	0.15	7.23	0.42	0.94	0.53	3.80	0.28	0.55	0.67
SD	0.02	0.01	0.01	0.01	0.02	0.02	0.11	0.03	0.30	0.11	1.22	0.027	0.12	0.10
r	0.710	0.292	0.165	-0.279	-0.726	0.480	-0.031	0.337	-0.570	0.622	0.301	0.745	-0.107	0.645
p-level	0.002	0.273	0.541	0.296	0.001	0.060	0.909	0.202	0.021	0.010	0.257	0.001	0.693	0.007

r, Correlation coefficient with the fibrosis stage; PME, Phosphomonoesters; Pi, Inorganic phosphate; PDE, Phosphodiesters; NTP, nucleoside triphosphate, pH_i, Intracellular pH.

*difference between the infected and the control group is significant at p<0.05.

**difference between the infected and the control group is significant at p<0.01.

https://doi.org/10.1371/journal.pntd.0005778.t003





https://doi.org/10.1371/journal.pntd.0005778.g005

NTP-γ to NTP-α as well as NTP-γ to NTP-β ratios did not differ, whereas NTP-β to NTP-α ratio markedly increased in the infected group. Moreover, NTP-β/NTP-α ratio correlated with the fibrosis stage (r = 0.622, p = 0.010), cholesterol content (r = 0.622, p = 0.005) and cholesterol to phospholipids ratio in the liver (r = 0.664, p = 0.005).

The calculated intracellular pH did not differ in the control and infected groups.

Western blot analysis

5'-AMP-activated protein kinase (AMPK) has been implicated in the control of hepatic glucose and lipid homeostasis, including fatty acid and sterol synthesis. It has been proposed to act as a 'metabolic master switch' mediating cellular adaptation to environmental or nutritional stress factors [49]. As illustrated in Fig 5 the amount of AMPK, phosphorylated by Thr172 residue of α -subunit (phospho-AMPK), decreased in *O. felineus* infected liver. This means that the activity of kinase in the infected liver is lower than that in the healthy one. Statistical analysis showed that the relative levels of phospho-APMK in the hamster liver correlated with the NTP- α level, determined according to 31P MRS, (r = 0.585, p = 0.028), cholesterol (r = -0.726, p = 0.003) and triglycerides (r = -0.590, p = 0.026) concentrations in serum as well as the fibrosis stage (r = -0.607, p = 0.021).

Discussion

O. felineus infection leads to significant pathological changes in the bile duct and liver of the hamsters, which results in chronic cholangitis and pronounced periductal/mild portal fibrosis. The key biochemical marker of hepatic parenchymal cell injury is the observed 3-fold increase in ALT in serum of the infected hamsters occurring due to the release of the enzymes from liver tissue into the circulation. The ALT level elevation in opisthorchiasis has been emphasized by several authors [50,51]. The observed hepatomegaly in the infected hamsters, a common sign of parasitic infection [52], is caused rather by the inflammatory process in the liver

[53,54]. The lowering of the protein concentration in liver tissue (per gram) of the infected animals may indicate an impaired protein synthesis function of hepatocytes. In the meantime, there is no difference in the total protein content calculated per whole liver between the infected and intact hamsters. Consequently, physiological adaptation to a metabolic demand as a driving mechanism of liver enlargement in the infected hamsters cannot be excluded [55].

O. felineus-induced liver lesions are clearly identified by MRI. The key MRI finding observed in our study is the pronounced intrahepatic biliary ductal dilatation with extension of T2-hyperintense damaged zone around the dilated bile duct. These spreading hyperintense zones correspond to fibrosis in adjacent to the bile duct hepatic tissue clearly identified by the histological analysis. The observed hyperintense signal from the bile duct wall is related to the inflammatory process diagnosed in the bile duct according to the histopathalogical study.

It is important to emphasize the difference of the *O. felineus* MRI-pattern from that of closely related liver fluke infections. Choi et al. [33] described that the hallmark of radiologic findings in clonorchiasis is diffuse, mild dilatation of the intrahepatic biliary tree, especially the peripheral bile ducts, without any evidence of obstruction. In our study the dilatation was sufficiently pronounced. Yet, MRI in humans may slightly differ from MRI in animal models due to differences in size of the organ. At the same time, the T2-weighted image of *O. felineus* infected liver demonstrated wider and more pronounced hyperintense zones in liver tissue around the dilated bile duct, compared to the *O. viverrini* infected liver [34]. The origin of these differences is probably related to the fact that *O. felineus* causes more pronounced liver damage, as opposed to *O. viverrini* [56]. Therefore, the development of an approach to differential diagnosis of *O. felineus* infection from other helminth infections requires further investigation in human population.

There is an additional reason for implementing of MRI for opisthorchiasis diagnosis. Since bile duct dilatation and T2-enchancement have taken place in different liver lobes of the infected individuals, liver biopsy sampling variability rises and limits predictive validity of biopsy analysis for staging liver fluke-induced fibrosis. In comparison with microscopic stool examination, MRI is a more expensive technique for diagnosing *O. felineus* infection, but it allows clinicians to evaluate opisthorchiasis-related liver damage [57]. Moreover, MRI can be very helpful for opisthorchiasis detection in infected people in non-endemic countries.

Morphological and functional damage of the liver that resulted from mechanical, inflammatory, and toxic injury may lead to disturbance of the metabolic processes in the liver.

Our 1H MRS data indicated no changes in the lipid profile of the liver and no fat droplets were observed in the liver of the infected hamsters according to histological examination. At the same time, biochemical analysis of hepatic tissue showed the elevated cholesterol concentration in liver tissue. Since MR techniques are sensitive only to molecules with a high degree of rotational molecular motion, the observed lipid resonances arise only from the intracellular fat droplets, whereas membrane lipids are not detectable using MRS [48]. Thus, the combination of MRS and biochemical data allows us to assume that cholesterol in the liver accumulates in the cellular membrane and the cholesterol to phospholipids ratio rises, respectively. This process may lead to loss of fluidity of liquid-crystalline domains in the cell membrane [58]. In turn, an increase in membrane rigidity may affect conformational freedom of integral membrane proteins and, consequently, their function. Therefore, the function of canalicular membrane transporters of cholesterol can be impaired even more, leading to enhancement of cholesterol breakdown [59].

Observed in our study hyperlipidemia in combination with the increased cholesterol concentration and elevated cholesterol to phospholipid ratio in liver tissue indicate the disturbance of lipid metabolism in *O. felineus* infection. Alteration of lipid metabolism in parasite infection is emphasized by several authors [60-62]. For example, rise in the cholesterol/ phospholipid ratio in liver plasma membrane and hyperlipidemia accompanied by strong accumulation of lipids in hepatic tissues during ancylostomiasis has been reported [60,63]. In case of *O. felineus* infection, cholesterol accumulates in the liver, however, no triglyceride and phospholipid accumulation occurs. It is little known about the main source and proportion of nutrients for *O. felineus* [37,64], but liver flukes inhabiting the bile ducts are likely able to uptake nutrients from the bile. Young et al. reported [65] that closely related species *O. viverrini* and *C. sinensis* transcribe genes encoding enzymes and accessory proteins directly involved in processing of bile constituents, and can utilize free lipids. Notably, these genes are absent in blood flukes and tapeworms.

Generally, cholestasis and biliary injury are associated with marked abnormalities of cholesterol metabolism and pronounced elevations in serum cholesterol and LDL [66,67]. Here mechanisms that can lead to cholesterol accumulation in liver tissue should be pointed out. At first, injury of epithelial bile duct cells can result in cholesterol excretion failure. Secondly, infection-associated cellular injury may need extra cholesterol for new membrane synthesis and be beneficial to the host for protection from harmful effects of the stimuli [68]. The last assumption has good concordance with our 31P MRS data. The increased PME level has been hypothesized to be associated with intensified cell membrane synthesis [27,69] and regeneration [70,71]. The observed in our study correlations of PME and NTP- α peaks with the ALT, AST, cholesterol, HDL levels in serum and the cholesterol/phospholipid ratio in liver tissue of the infected hamsters supported the relation of these spectral parameters to liver injury and fibrosis.

However, there are some doubts concerning the power of the 31P MRS technique for assessing the degree of hepatic fibrosis [72]. Published literature has significant variations in the discovered correlation between 31P MRS peaks or its ratios and various liver abnormalities. Nonetheless a fibrosis marker, emphasized by the majority of authors, is the PME/PDE ratio; changes in the PME/PDE are thought to be associated with an increase in the regenerative efforts made by the damaged liver [73]. In our study we found neither changes in this ratio in *O. felineus* infection nor its correlation with fibrosis [72].

Thus, the correlation of NTP- α and PME peaks with fibrosis observed in our study can mainly reflect the metabolic changes in the liver, i.e. intensification of gluconeogenesis [74–76]. The assumption about intensification of gluconeogenesis in the liver in *O. felineus* infection is supported by the rise in PME as well as PME/NTP- α , PME/NTP- γ ratios in the infected group of hamsters. PME resonance, apart from PC and PE (that are believed to represent phospholipid cell membrane precursors), also includes gluconeogenesis intermediates, such as glucose-6-phosphate (G6P) and 3-phosphoglycerate (3PG) [27,69,74].

It is essential to describe the origin of the metabolic alteration. The critical role in local regulation of metabolism is played by the energetic status of the liver, especially the level and proportion of high-energy phosphates. In 31P MRS spectra, NTP- γ , - β and - α peaks are attributed to ATP, but also contain resonances from other triphosphates [77]. The NTP- β peak contains information almost exclusively from ATP [78]. In our experiment, the level of NTP- β as well as the Pi/NTP- β ratio in the liver of the infected hamsters did not change. Note that the ATP concentration in the liver of *O. felineus* infected animals, determined by the chemiluminescent assay, also maintained at a constant level. Thus, it may be concluded, that at this stage of infection the bioenergetics of the liver is not significantly disturbed.

Nonetheless, NTP- α and- γ peaks, apart from ATP, contain a contribution from ADP [77] in turn NAD+/NADP+ and NADH/NADPH resonances are the components of the NTP- α peak. 31P MRS data showed that Pi/NTP- α and Pi/NTP- γ as well as NTP- β /NTP- α ratios increased markedly in the infected group. Although 31P MRS did not allow us to make an accurate conclusion about the level of AMP and ADP, the observed spectral alteration indicated the changes in the high-energy phosphates ratio in the liver of the infected hamsters. In turn, this proportion plays a crucial role in cell metabolism regulation [79]. A drop in the AMP/ATP ratio results in dephosphorylation and inactivation of AMPK [49,80,81].

The decrease in the phospho-AMPK level in *O. felineus* infected liver and the correlation of the relative phospho-AMPK level with NTP- α peak intensity observed according to western blot analysis support our above-mentioned suggestions. So, when the AMP level is low and the ATP level is normal, glycolysis is nearly switched off and gluconeogenesis is promoted [82]. In turn, acetyl-CoA in the presence of adequate stores of ATP and low AMP levels is diverted to lipid synthesis [83]. Thus, lowering of the AMPK activity leads to a decrease in HMG-CoA reductase phosphorylation and promotes cholesterol synthesis [49]. Since acetyl-CoA carboxylase (ACC) is a substrate of AMPK, depletion of AMP increases the catalytic activity of ACC and the level of malonyl-CoA, which is a critical precursor for biosynthesis of fatty acids [49,80]. The invert correlations of the phospho-AMPK level with triglycerides and cholesterol serum concentration can be considered as an indirect evidence of lipid synthesis intensification in the infected animal liver. The mechanism underlying the decrease in the AMPK activity has to be further investigated, but it is important to note that, for example, TNF- α and IL-6 cytokines suppress AMPK [84,85].

At the system level, chronic liver injury and inflammation per se provoke chronic stress followed by a rise in the cortisol/corticosterone level. There are some publications about elevation of plasma cortisol in animals infected by parasitic worms [86,87]. This glucocorticoid(-s) has several effects on metabolism, in particular, it provokes mobilization of non-hexose substrates from extrahepatic tissues and stimulates synthesis of glucose in the liver. Simultaneously, chronic glucocorticoid excess is not associated with increased lipolysis in liver tissue; moreover, it may cause activation of hepatic lipogenesis [88,89]. What is more, determined in our study glycogen concentration in liver tissue of the experimental hamsters did not reduce during infection. Additionally, glucocorticoids demonstrate anti-inflammatory and immunosuppressive properties that can contribute to success of longstanding invasion of liver flukes. The contribution of cortisol/corticosterone to the observed metabolic disorder should be investigated further.

Apart from injury, inflammation and regeneration, the specific parasite-host interaction, i.e. molecules excreted by parasite adult worms during infection, may be the factor(s) leading to the observed metabolic changes [60,90]. In fact, the parasite's goal is to take over the host, to provide comfortable conditions for its own habitat.

Arguably, *Opisthorchiidae* flukes are unable to synthesize cholesterol *de novo*. As an alternative, for example, *O. viverrini* has molecular pathways to acquire, transport and process cholesterol from external lipids [65]. Thus, biosynthesis of cholesterol by the host is very important for the parasite. Here the question arises, whether the parasite may evolve some opportunity (excreted molecular stimuli) to regulate the cholesterol metabolism of the host. Indeed, Adam et al. observed marked proliferation of smooth endoplasmic reticulum (ER) in hepatocytes of animals infected with *O. viverrini* [91]. In turn, enzymes in smooth ER are involved in lipid synthesis, including phospholipids and steroids.

For energy production the *Opisthorchiidae* fluke uses anaerobic and aerobic glycolysis [65]; during anaerobic glycolysis glucose is catabolized most likely to lactate. In spite of the fact that the parasite's genome encodes all of the tricarboxylic acid cycle enzymes living in microaerobic environment, it cannot effectively metabolize end-products of glycolysis along oxidative pathways and, therefore, they have to be excreted into the surrounding tissues of the host [92]. Further end-products of glycolysis may be utilized by the host liver in gluconeogenesis (preferable in the liver) or oxidative energy producing pathways. Recently in *in vitro* experiments it has been shown that excretory products of *O. viverrini* altered glycolysis/gluconeogenesis in

normal human cholangiocytes (H69 cell line) [93]. The proposed changes of the high-energy phosphate ratios, as well as glucose metabolism, particularly, the gluconeogenesis rate, in *O. felineus* infected liver require additional investigation. However, the hypothesis about prevention of AMPK signaling activation by the parasite to retain the biosynthetic function of the liver, for the organ to produce cholesterol and glucose that are essential for liver fluke survival, seems not incurious.

Having analyzed the whole set of the obtained data, we may conclude that *O. felineus* infection disturbs hepatic metabolism of the host. The liver is considered to be a metabolic power station of mammalians, where cholesterol homeostasis relies on an intricate network of cellular processes, which deregulations can lead to severe pathologies [94]. In turn, the decreased AMPK activation, indicated in the infected hamster liver, is known to be implicated in metabolic disorders and associated with cancer risk [95]. Future investigations may shed light on the mechanism that leads to the observed metabolic changes and evaluate the contribution of *O. felineus* infection to known metabolic disorders. The knowledge about the capabilities of the parasite to regulate metabolism of the host may be applied for seeking a target for anti-helminth therapy and serve as the basis for developing new approaches to human metabolic disease regulation.

Conclusion

The use of MR-techniques for investigating *O. felineus* infected liver has some advantages. MRI and MRS provide valuable information about the severity of liver impairments induced by opisthorchiasis. MR images of *O. felineus* infected liver have a characteristic pattern that differs from that of other liver fluke infections. The MRS findings point out the metabolic changes in the infected liver and identify the areas in which further scientific investigation is required. Further MR examination of infected humans allows clinicians to implement these methods in routine diagnosis of opisthorchiasis.

Supporting information

S1 Text. Blood and serum analysis. (DOCX)

S1 Table. Laboratory parameters of the study cohort. (DOCX)

S2 Table. Correlations between hepatic phosphorylated metabolites (³¹P MRS) and biochemical data of serum and liver tissue analysis. (DOCX)

Acknowledgments

The authors thank Professor Leonid N. Maslov and Ekaterina S. Prokudina for help with ATP assay.

Author Contributions

Conceptualization: Alexandra G. Pershina, Vladimir V. Ivanov.

Data curation: Alexandra G. Pershina.

Formal analysis: Alexandra G. Pershina.

Funding acquisition: Ludmila M. Ogorodova.

Investigation: Alexandra G. Pershina, Vladimir V. Ivanov, Lina V. Efimova, Oleg B. Shevelev, Sergey V. Vtorushin, Tatjana V. Perevozchikova.

Methodology: Alexandra G. Pershina, Vladimir V. Ivanov, Lina V. Efimova, Oleg B. Shevelev.

Project administration: Alexandra G. Pershina, Alexey E. Sazonov.

Resources: Alexey E. Sazonov, Ludmila M. Ogorodova.

Supervision: Ludmila M. Ogorodova.

Validation: Alexandra G. Pershina.

Writing - original draft: Alexandra G. Pershina.

Writing - review & editing: Alexandra G. Pershina.

References

- Sripa B, Mairiang E, Thinkhamrop B, Laha T, Kaewkes S, et al. (2009) Advanced periductal fibrosis from infection with the carcinogenic human liver fluke Opisthorchis viverrini correlates with elevated levels of IL-6. Hepatology 50: 1273–1281. https://doi.org/10.1002/hep.23134 PMID: 19676135
- Charoensuk L, Pinlaor P, Laothong U, Yongvanit P, Pairojkul C, et al. (2014) Bile canalicular changes and defective bile secretion in Opisthorchisviverrini-infected hamsters. Folia Parasitol (Praha) 61: 512– 522.
- 3. Mordvinov VA, Furman DP (2010) The Digenea parasite Opisthorchis felineus: a target for the discovery and development of novel drugs. Infect Disord Drug Targets 10: 385–401. PMID: 20701571
- 4. IARC Monogr Eval Carcinog Risks Hum. (1994) Infection with liver flukes (Opisthorchis viverrini, Opisthorchis felineus and Clonorchis sinensis). 61: 121–175.
- Xia J, Jiang S-c, Peng H-J (2015) Association between Liver Fluke Infection and Hepatobiliary Pathological Changes: A Systematic Review and Meta-Analysis. PLoS ONE 10: e0132673. https://doi.org/ 10.1371/journal.pone.0132673 PMID: 26186510
- Pakharukova MY, Mordvinov VA, 2016 (2016) The liver fluke Opisthorchis felineus: biology, epidemiology and carcinogenic potential. Trans R Soc Trop Med Hyg 110: 28–36. https://doi.org/10.1093/trstmh/ trv085 PMID: 26740360
- Fedorova OS, Kovshirina YV, Kovshirina AE, Fedotova MM, Deev IA, et al. (2016) [Analysis of opisthorchis felineus infection and liver and intrahepatic bile ducts cancer incidence rate in Russian Federation][Article in Russian]. Bulletin of Siberian Medicine 15: 147–158
- 8. Maksimova GA, Zhukova NA, Kashina EV, Lvova MN, Katokhin AV, et al. (2015) [Role of opisthorchis felineus on induction of bile duct cancer][Article in Russian]. Parazitologiia 49: 3–11. PMID: 26016330
- Prakobwong S, Yongvanit P, Hiraku Y, Pairojkul C, Sithithaworn P, et al. (2010) Involvement of MMP-9 in peribiliaryfibrosis and cholangiocarcinogenesis via Rac1-dependent DNA damage in ahamster model. Int J Cancer 127: 2576–2587. https://doi.org/10.1002/ijc.25266 PMID: 20162672
- Yongvanit P, Pinlaor S, Bartsch H (2012) Oxidative and nitrative DNA damage: keyevents in opisthorchiasis-induced carcinogenesis. Parasitol Int 61: 130–135. https://doi.org/10.1016/j.parint.2011.06.011 PMID: 21704729
- Lurie Y, Webb M, Cytter-Kuint R, Shteingart S, Lederkremer GZ (2015) Non-invasive diagnosis of liver fibrosis and cirrhosis World J Gastroenterol 21: 11567–11583. https://doi.org/10.3748/wjg.v21.i41. 11567 PMID: 26556987
- 12. Baranova A, Lal P, Birerdinc A, Younossi ZM (2011) Non-Invasive markers for hepatic fibrosis. BMC Gastroenterology 11: 91. https://doi.org/10.1186/1471-230X-11-91 PMID: 21849046
- 13. Bataller R, Brenner DA (2005) Liver fibrosis. J Clin Invest 115: 209–218. https://doi.org/10.1172/ JCI24282 PMID: 15690074
- 14. Ferrell L (2000) Liver Pathology: Cirrhosis, Hepatitis, and Primary Liver Tumors. Update and Diagnostic Problems. Mod Pathol 13: 679–704. https://doi.org/10.1038/modpathol.3880119 PMID: 10874674
- Cherdron A, Fiegel P (1992) [Opisthorchis felineus—the cat liver fluke. Differential diagnosis of rightside upper abdominal pain]. Dtsch Med Wochenschr [Article in German] 117: 328–331. <u>https://doi.org/ 10.1055/s-2008-1062315 PMID: 1544352</u>
- **16.** Pozio E, Armignacco O, Ferri F, Morales MAG (2013) Opisthorchis felineus, an emerging infection in Italy and its implication for the European Union. ActaTropica 126: 54–62.

- Melling N, Hohenberger W, Yedibela SI (2009) Opisthorchiasis mimicking primary biliary cirrhosis as an indication for liver transplantationq. Journal of Hepatology 50: 1057–1059. https://doi.org/10.1016/j. jhep.2008.11.029 PMID: 19329217
- Armignacco O, Ferri F, Gomez-Morales MA, Caterini L, Pozio E (2013) Cryptic and Asymptomatic Opisthorchis felineus Infections. Am J Trop Med Hyg 88: 364–366. https://doi.org/10.4269/ajtmh.2012. 12-0280 PMID: 23249682
- Tan WB, Shelat VG, Diddapur RK (2009) Oriental liver fluke infestation presenting more than 50 years after immigration. Ann Acad Med Singapore 38: 735–736. PMID: 19736582
- 20. Castera L, Pinzani M (2010) Non-invasive assessment of liver fibrosis: are we ready? Lancet 375: 1419–1420. https://doi.org/10.1016/S0140-6736(09)62195-4 PMID: 20417845
- Faria SC, Ganesan K, Mwangi I, Shiehmorteza M, Viamonte B, et al. (2009) MR imaging of liver fibrosis: current state of the art. Radiographics 29: 1615–1635. <u>https://doi.org/10.1148/rg.296095512</u> PMID: 19959511
- 22. (1989) NMR in Biomedicine. The Physical Basis; Fukushima E, editor: AIP-Press. 180 p.
- Taouli B, Koh DM (2010) Diffusion-weighted MR imaging of the liver. Radiology 254: 47–66. <u>https://doi.org/10.1148/radiol.09090021</u> PMID: 20032142
- 24. Patel HT, Shah AJ, Khandelwal SR, Patel HF, Patel MD (2009) MR Cholangiopancreatography at 3.0 T1. Radiographics 29: 1689–1706. https://doi.org/10.1148/rg.296095505 PMID: 19959516
- Hwang J-H, Choi CS (2015) Use of in vivo magnetic resonance spectroscopy for studying metabolic diseases. Exp Mol Med 47: e139. https://doi.org/10.1038/emm.2014.101 PMID: 25656949
- Qayyum A (2009) MR spectroscopy of the liver: principles and clinical applications. Radiographics 29: 1653–1664. https://doi.org/10.1148/rg.296095520 PMID: 19959513
- Fischbach F, Schirmer T, Thormann M, Freund T, Ricke J, et al. (2008) Quantitative proton magnetic resonance spectroscopy of the normal liver and malignant hepatic lesions at 3.0 Tesla. Eur Radiol 18: 2549–2558. https://doi.org/10.1007/s00330-008-1040-8 PMID: 18491103
- Longo R, Pollesello P, Ricci C, Masutti F, Kvam BJ, et al. (1995) Proton MR spectroscopy in quantitative in vivo determination of fat content in human liver steatosis. J Magn Reson Imaging 5: 281–285. PMID: 7633104
- Helzberg JH, Brown MS, Smith DJ, Gore JC, Gordon ER (1987) Metabolic state of the rat liver with ethanol: Comparison of in vivo31 phosphorus nuclear magnetic resonance spectroscopy with freeze clamp assessment. Hepatology 7: 83–88. PMID: <u>3804209</u>
- Jalan R, Sargentoni J, Coutts GA, Bell JD, Rolles K, et al. (1996) Hepatic phosphorus-31 magnetic resonance spectroscopy in primary biliary cirrhosis and its relation to prognostic models. Gut 39: 141–146. PMID: 8881826
- Roelsgaard K, StBdkilde-Jmgensen H, Dmstrup S, Djurhuust JC (1993) Non-invasive Investigation of Parenchymal Liver Disease Using 31PN MR Spectroscopy. NMR IN BIOMEDICINE 6: 383–388.
 PMID: 8148234
- 32. Choi BI, Kim TK, Han JK (1998) MRI of clonorchiasis and cholangiocarcinoma. J Magn Reson Imaging 8: 359–366. PMID: 9562062
- Choi D, Hong S-T (2007) Imaging diagnosis of clonorchiasis. Korean J Parasitol 45: 77–85. https://doi. org/10.3347/kjp.2007.45.2.77 PMID: 17570969
- Hanpanich P, Pinlaor S, Charoensuk L, Yongvanit P, Thomas C, et al. (2013) MRI and 1H MRS evaluation for the serial bile duct changes in hamsters after infection with Opisthorchis viverrini. Magn Res Imag 31: 1418–1425.
- Jaruchainiwat S, Hanpanich P, Sereerak P, Upontain S, Tangkawattana P, et al. (2015) Potentiality of 3T MRI for the demonstration of biliary changes in Opisthorchis viverrini infected cats. Congress ECR 2015.
- **36.** (2011) Guide for the care and use of laboratory animals (Eighth Edition). Washington, D.C.: The National Academies press. 220 p.
- **37.** Pershina AG, Saltykova IV, Ivanov VV, Perina EA, Demin AM, et al. (2015) Hemozoin "knobs" in Opisthorchis felineus infected liver. Parasites & Vectors 8: 459.
- Rata M, Giles SL, deSouza NM, Leach MO, Payne GS (2014) Comparison of three reference methods for the measurement of intracellular pH using 31P MRS in healthy volunteers and patients with lymphoma. NMR Biomed 27: 158–162. https://doi.org/10.1002/nbm.3047 PMID: 24738141
- Bedossa P, Poynard T (1996) An algorithm for the grading of activity in chronic hepatitis C. The META-VIR Cooperative Study Group. Hepatology 24: 289–293. https://doi.org/10.1002/hep.510240201 PMID: 8690394

- Brunt EM, Janney CG, Di Bisceglie AM, Neuschwander-Tetri BA, Bacon BR (1999) Nonalcoholic steatohepatitis: a proposal for grading and staging the histological lesions. Am J Gastroenterol 94: 2467– 2474. https://doi.org/10.1111/j.1572-0241.1999.01377.x PMID: 10484010
- Folch J, Less M, Sloane Stanley GH (1957) A simple method for the isolation and purification of total lipides from animal tissues. J Biol Chem 226: 497–509. PMID: 13428781
- Bradford MM (1976) A rapid and sensitive method for the quantitation of microgram quantities of protein utilizing the principle of protein-dye binding. Anal Biochem 72: 248–254. PMID: 942051
- Passonneau JV, Lauderdale VR (1974) A comparison of three methods of glycogen measurement in tissues. Analytical Biochemistry 60: 405–441. PMID: 4844560
- Laemmli UK (1970) Cleavage of structural proteins during the assembly of the head of bacteriophage T4. Nature 227: 680–685. PMID: 5432063
- Delikatny EJ, Chawla S, Leung D-J, Poptani H (2011) MR-Visible Lipids and the Tumor Microenvironment. NMR Biomed 24: 592–611. https://doi.org/10.1002/nbm.1661 PMID: 21538631
- 46. Strobel K, van den Hoff J, Pietzsch J (2008) Localized proton magnetic resonance spectroscopy of lipids in adipose tissue at high spatial resolution in mice in vivo. J Lipid Res 49: 473–480. https://doi.org/ 10.1194/jlr.D700024-JLR200 PMID: 18024705
- Cheung JS, Fan SJ, Gao DS, Chow AM, Yang J, et al. (2011) In vivo lipid profiling using proton magnetic resonance spectroscopy in an experimental liver fibrosis model. Acad Radiol 18: 377–383. https://doi.org/10.1016/j.acra.2010.10.012 PMID: 21167757
- Corbin IR, Furth EE, Pickup S, Siegelman ES, Delikatny EJ (2009) In vivo assessment of hepatic triglycerides in murine non-alcoholic fatty liver disease using magnetic resonance spectroscopy. Biochim Biophys Acta 1791: 757–763. https://doi.org/10.1016/j.bbalip.2009.02.014 PMID: 19269347
- 49. Viollet B, Foretz M, Guigas B, Horman S, Dentin R, et al. (2006) Activation of AMP-activated protein kinase in the liver: a new strategy for the management of metabolic hepatic disorders. J Physiol 574: 41–53. https://doi.org/10.1113/jphysiol.2006.108506 PMID: 16644802
- Boonjaraspinyo S, Boonmars T, Aromdee C, Srisawangwong T, Kaewsamut B, et al. (2009) Turmeric reduces inflammatory cells in hamster opisthorchiasis. Parasitol Res 105: 1459–1463. https://doi.org/ 10.1007/s00436-009-1553-3 PMID: 19633869
- Maksimova GA, Pakharukova MY, Kashina EV, Zhukova NA, Lvova MN, et al. (2016) The morphofunctional and biochemical characteristics of opisthorchiasis-associated cholangiocarcinoma in a Syrian hamster model. Russ J Genet Appl Res 6: 454.
- Wang Y (2017) Pathophysiology of Parasitic Disease. In: Hongjun LI, editor. Radiology of Parasitic Diseases A Practical Approach: Springer Netherlands. pp. 9–11.
- 53. Wilson S, Jones FM, Mwatha JK, Kimani G, Booth M, et al. (2009) Hepatosplenomegaly associated with chronic malaria exposure: evidence for a pro-inflammatory mechanism exacerbated by schistosomiasis. Parasite Immunol 31: 64–71. https://doi.org/10.1111/j.1365-3024.2008.01078.x PMID: 19149774
- Walker WA, Mathis RK (1975) Hepatomegaly. An approach to differential diagnosis. Pediatr Clin North Am 22: 929–942. PMID: <u>1105367</u>
- Matsuo T, Izumori K (2006) Effects of dietary D-psicose on diurnal variation in plasma glucose and insulin concentrations of rats. Biosci Biotechnol Biochem 70: 2081–2085. PMID: <u>16960391</u>
- 56. Lvova MN, Tangkawattana S, Balthaisong S, Katokhin AV, Mordvinov VA, et al. (2012) Comparative histopathology of Opisthorchisfelineus and Opisthorchis viverrini in a hamster model: An implication of high pathogenicity of the European liver fluke. Parasitology International 61: 167–172. https://doi.org/ 10.1016/j.parint.2011.08.005 PMID: 21854870
- Demin AM, Pershina AG, Ivanov VV, Nevskaya KV, Shevelev OB, et al. (2016) 3-Aminopropylsilanemodified iron oxide nanoparticles for contrast-enhanced magnetic resonance imaging of liver lesions induced by Opisthorchis felineus. Int J Nanomedicine 11: 4451–4463. <u>https://doi.org/10.2147/IJN.</u> S111880 PMID: 27660439
- Cooper RA (1978) Influence of increased membrane cholesterol on membrane fluidity and cell function in human red blood cells. J Supramol Struct 8: 413–430. <u>https://doi.org/10.1002/jss.400080404</u> PMID: 723275
- 59. Boyer JL (2013) Bile Formation and Secretion. Compr Physiol 3: 1035–1078. https://doi.org/10.1002/ cphy.c120027 PMID: 23897680
- Mukerjee S, C R., Tekwani BL, Gupta S, Katiyar JC, et al. (1990) Molecular basis of hyperlipidemia in golden hamsters during experimental infection with Ancylostoma ceylanicum (Nematoda:Strongylidae). Int J Parasitol 20: 217–223. PMID: 2332280
- Stanley RG, Jackson CL, Griffiths K, Doenhoff MJ (2009) Effects of Schistosoma mansoni worms and eggs on circulating cholesterol and liver lipids in mice. Atherosclerosis 207: 131–138. <u>https://doi.org/ 10.1016/j.atherosclerosis.2009.04.037</u> PMID: 19464685

- **62.** Tsuboi T, Hirai K (1986) Lipid metabolism in golden hamsters infected with plerocercoids of Spirometra erinacei (Cestoda: Pseudophyllidea). Parasitology International 93: 143–151.
- **63.** Srivastava SC (1994) Liver plasma membrane-bound enzymes and lipids in golden hamsters infected with Ancylostoma ceylanicum. Int J Parasitol 24: 249–251. PMID: 7913086
- **64.** Lvova M, Zhukova M, Kiseleva E, Mayboroda O, Hensbergen P, et al. (2016) Hemozoin is a product of heme detoxification in the gut of the most medically important species of the family Opisthorchiidae. Int J Parasitol 46: 147–156. https://doi.org/10.1016/j.ijpara.2015.12.003 PMID: 26812025
- 65. Young ND, Nagarajan N, Lin SJ, Korhonen PK, Jex AR, et al. (2014) The Opisthorchis viverrini genome provides insights into life in the bile duct. Nat Commun 5: 4378. https://doi.org/10.1038/ncomms5378 PMID: 25007141
- Longo M, Crosignani A, Podda M (2001) Hyperlipidemia in Chronic Cholestatic Liver Disease. Curr Treat Options Gastroenterol 4: 111–114. PMID: <u>11469968</u>
- Parl F, Gutstein WH, D'Aguillo AF, Baez A (1975) Endothelial injury. Association with elevations of serum bile acid and cholesterol concentration in biliary-obstructed rats. Atherosclerosis 21: 135–146. PMID: 1131304
- 68. Laothong U, Pinlaor P, Boonsiri P, Hiraku Y, Khoontawad J, et al. (2013) α-Tocopherol and lipid profiles in plasma and the expression of α-tocopherol-related molecules in the liver of Opisthorchis viverriniinfected hamsters. Parasitol Int 62: 127–133. <u>https://doi.org/10.1016/j.parint.2012.11.002</u> PMID: 23178928
- Corbin IR, Buist R, Peeling J, Zhang M, Uhanova J, et al. (2003) Hepatic 31P MRS in rat models of chronic liver disease: assessing the extent and progression of disease. Gut 1046–1053. PMID: 12801965
- 70. Solga SF, Horska A, Clark JM, Diehl AM (2005) Hepatic 31P magnetic resonance spectroscopy: a hepatologist's user guide. Liver Int 25: 490–500. <u>https://doi.org/10.1111/j.1478-3231.2005.01131.x</u> PMID: 15910484
- Orlacchio A, Bolacchi F, Angelico M, Mancini A, Cozzolino V, et al. (2008) In vivo, high-field, 3-Tesla 1H MR spectroscopic assessment of liver fibrosis in HCV-correlated chronic liver disease. Radiol med 113: 289–299. https://doi.org/10.1007/s11547-008-0239-8 PMID: 18386129
- 72. Godfrey EM, Patterson AJ, Priest AN, Davies SE, Joubert I, et al. (2012) A comparison of MR elastography and 31P MR spectroscopy with histological staging of liver fibrosis. Eur Radiol 22: 2790–2797. https://doi.org/10.1007/s00330-012-2527-x PMID: 22752441
- **73.** Taylor-Robinson SD (2001) Applications of magnetic resonance spectroscopy to chronic liver disease. Clin Med (Lond) 1:54–60.
- 74. Sharma R, Sinha S, Danishad KA, Vikram NK, Gupta A, et al. (2009) Investigation of hepatic gluconeogenesis pathway in non-diabetic Asian Indians with non-alcoholic fatty liver disease using in vivo ((31)P) phosphorus magnetic resonance spectroscopy. Atherosclerosis 203: 291–297. <u>https://doi.org/10. 1016/j.atherosclerosis.2008.06.016</u> PMID: 18674764
- 75. Leij-Halfwerk S, Dagnelie PC, Van Den Berg JW, Wilson JH, Sijens PE (2000) Hepatic sugar phosphate levels reflect gluconeogenesis in lung cancer: simultaneous turnover measurements and 31P magnetic resonance spectroscopy in vivo. Clin Sci (Lond) 98: 167–174.
- 76. Changani KK, Jalan R, Cox IJ, Ala-Korpela M, Bhakoo K, et al. (2001) Evidence for altered hepatic gluconeogenesis in patients with cirrhosis using in vivo 31-phosphorus magnetic resonance spectroscopy. Gut 49: 557–564. https://doi.org/10.1136/gut.49.4.557 PMID: 11559655
- 77. Iles RA, Stevens AN, Griffiths JR, Morris PG (1985) Phosphorylation status of liver by 31P-n.m.r. spectroscopy, and its implications for metabolic control. A comparison of 31P-n.m.r. spectroscopy (in vivo and in vitro) with chemical and enzymic determinations of ATP, ADP and Pi. Biochem J 229: 141–151. PMID: 4038253
- Davidson BR, Barnard ML, Changani KK, Taylor-Robinson SD (1997) Liver transplantation: current and potential applications of magnetic resonance spectroscopy. Liver Transpl Surg 3: 481–493. PMID: 9346790
- 79. Dzeja P, Terzic A (2009) Adenylate Kinase and AMP Signaling Networks: Metabolic Monitoring, Signal Communication and Body Energy Sensing. Int J Mol Sci 10: 1729–1772. <u>https://doi.org/10.3390/jjms10041729 PMID: 19468337</u>
- **80.** Farooqui AA (2013) Metabolic Syndrome. An Important Risk Factor for Stroke, Alzheimer Disease, and Depression. New York: Springer Science+Business Media.
- Lane MD, Cha SH (2009) Effect of glucose and fructose on food intake via malonyl-CoA signaling in the brain. Biochemical and Biophysical Research Communications 382: 1–5. https://doi.org/10.1016/j. bbrc.2009.02.145 PMID: 19265677
- 82. Berg JM, Tymoczko JL, Stryer L (2002) Biochemistry. New York: W H Freeman.

- Chatterjea MN, Shinde R (2012) Textbook of medical biochemistry. London: New Delhi: Jaypee Brothers Medical Publications (P) Ltd. 876 p.
- Steinberg GR, Michell BJ, van Denderen BJ, Watt MJ, Carey AL, et al. (2006) Tumor necrosis factor alpha-induced skeletal muscle insulin resistance involves suppression of AMP-kinase signaling. Cell Metab 4: 465–474. https://doi.org/10.1016/j.cmet.2006.11.005 PMID: 17141630
- Ko HJ, Zhang Z, Jung DY, Jun JY, Ma Z, et al. (2009) Nutrient stress activates inflammation and reduces glucose metabolism by suppressing AMP-activated protein kinase in the heart. Diabetes 58: 2536–2546. https://doi.org/10.2337/db08-1361 PMID: 19690060
- Prichard RK, Hennessy DR, Griffiths DA (1974) Endocrine responses of sheep to infection with Trichostrongylus colubriformis. Res Vet Sci 17: 182–187. PMID: 4529022
- Edith R, Thilagar MB, Godara R, Sharma RL (2010) Tropical liver fluke-induced stress in experimentally infected and immunised buffaloes. Vet Rec 167: 571–575. <u>https://doi.org/10.1136/vr.c4523</u> PMID: 21257419
- Fraser R, Ingram MC, Anderson NH, Morrison C, Davies E, et al. (1999) Cortisol effects on body mass, blood pressure, and cholesterol in the general population. Hypertension 33: 1364–1368. PMID: 10373217
- Macfarlane DP, Forbes S, Walker BR (2008) Glucocorticoids and fatty acid metabolism in humans: fuelling fat redistribution in the metabolic syndrome. J Endocrinol 197: 189–204. <u>https://doi.org/10.1677/ JOE-08-0054 PMID: 18434349</u>
- 90. Thompson SN (1991) Applications of nuclear magnetic resonance in parasitology. J Parasitol 77: 1–20. PMID: 1992079
- Adam R, Hinz E, Sithithaworn P, Pipitgool V, Storch V (1993) Ultrastructural hepatic alterations in hamsters and jirds after experimental infection with the liver fluke Opisthorchis viverrini. Parasitol Res 79: 357–364. PMID: 8415540
- Corbin I, Blackburn BJ, Novak M (1999) Metabolism of [3-13C]alanine in liver of mice infected with cysticerci of Taenia crassiceps. Canadian Journal of Zoology 77: 1367–1372.
- 93. Chaiyadet S, Smout M, Laha T, Sripa B, Loukas A, et al. (2016) Proteomic characterization of the internalization of Opisthorchis viverrini excretory/secretory products in human cells. Parasitol Int:
- 94. Trapani L, Segatto M, Pallottini V (2012) Regulation and deregulation of cholesterol homeostasis: The liver as a metabolic "power station". World J Hepatol 4: 184–190. <u>https://doi.org/10.4254/wjh.v4.i6.184</u> PMID: 22761969
- **95.** Fay JR, Steele V, Crowell JA (2009) Energy homeostasis and cancer prevention: the AMP-activated protein kinase. Cancer Prev Res (Phila) 2: 301–309.